Author's response to reviews

Title: Botulinum toxin in gastric submucosa reduces stimulated HCl production in rats.

Authors:

Matteo Runfola (runfola@tiscali.it)
Dr Simone Rossi (sinodoc@tiscali.it)
Simona Panunzi (simona.panunzi@biomatematica.it)
Pier Luigi Spada (plspada@hotmail.com)
Daniele Gui (daniele.gui@rm.unicatt.it)

Version: 4
Date: 22 Aug 2003

No other reviewer's reports were present.

Editor's comments:

* Qualifications - Please remove the authors' qualifications (e.g. PhD) from the title page

Done

* References - Please insert a space between the text and the cited reference number

Done

> * References - Citations in the reference list should not contain the > suffix et al but list all named authors, regardless of how many there > are.

Done

* Figures - It is important for the final layout of the manuscript, that the figures are cropped as closely as possible to minimise white space around the image. Our online figure cropping guide may be of some help, and can be viewed here: http://www.biomedcentral.com/info/ifora/cropping

Done